Author’s response to reviews

Title: Burden of influenza, healthcare seeking behaviour and hygiene measures during the A(H1N1)2009 pandemic in France: a population based study

Authors:

Dieter Van Cauteren (d.vancauteren@invs.sante.fr)
Sophie Vaux (s.vaux@invs.sante.fr)
Henriette de Valk (h.devalk@invs.sante.fr)
Yann Le Strat (y.lestrat@invs.sante.fra)
Véronique Vaillant (v.vaillant@invs.sante.fr)
Daniel Lévy-Bruhl (d.levybruhl@invs.sante.fr)

Version: 3 Date: 9 October 2012

Author’s response to reviews:

Dear Ms Audrey Ann Reyes and Dr. Danielle Ompad,

Please find enclosed the manuscript referenced above, which has been revised taking into account the final comment of the reviewers. We would like to submit this revised paper for publication as a research article in BMC Public Health.

As you will see, the sentence “The sample was representative in terms of gender, region, town size, and age (with the exception that children under five years of age were overrepresented due to the survey method)” has been added to the manuscript in the Results section.

We hope that this manuscript will satisfy the reviewers and look forward seeing the paper published in BMC Public Health.

Thank you for your consideration.

Yours sincerely,

Sophie Vaux, PharmD